% Please use the skeleton file you have received in the 
% invitation-to-submit email, where your data are already
% filled in. Otherwise please make sure you insert your 
% data according to the instructions in PoSauthmanual.pdf
%\documentclass[draft]{PoS}
\documentclass{PoS}
%\usepackage{ngerman}
\usepackage{slashed}
\newcommand{\rd}{\mathrm{d}}
\newcommand{\PW}{\mathrm{W}}
\newcommand{\PZ}{\mathrm{Z}}
\newcommand{\Pg}{\mathrm{g}}
\newcommand{\alps}{\alpha_{\mathrm{s}}}
\newcommand{\rT}{\mathrm{T}}
\newcommand{\Or}{\mathcal{O}}

\title{Vector-boson production at the LHC:\\
 QCD and electroweak effects}

\ShortTitle{Vector-boson production at the LHC}

\author{\speaker{Tobias Kasprzik}\\
        Institut f\"ur Theoretische Teilchenphysik (TTP), \\
        Karlsruher Institut f\"ur Technologie (KIT)\\
        E-mail: \email{kasprzik@particle.uni-karlsruhe.de}}

%\author{Another Author\\
%        Affiliation\\
%        E-mail: \email{...}}

      \abstract{A profound understanding of vector-boson production
        processes is of crucial importance at the LHC. The corresponding
        cross sections are large, and the final states are easy to
        reconstruct due to the clean signatures in the leptonic decay
        modes. Therefore, such processes play an important role as
        backgrounds in a large variety of new-physics signals, and they
        may furthermore help to better understand the well-established
        Standard-Model physics in a hadron-collider environment.  We
        review the recent progress in the theoretical description of
        higher-order QCD and electroweak effects in vector-boson
        production at the LHC, discussing the Drell--Yan
        process, vector-boson pair production and vector-boson
        production with associated jets.\\
\\
\\
{\it TTP11-27 \\
     SFB/CPP-11-57\\
     LPN11-59}
}

\FullConference{The 2011 Europhysics Conference on High Energy Physics, EPS-HEP
2011,\\
		July 21-27, 2011\\
		Grenoble, Rh\^one-Alpes, France}

\begin{document}
\begin{thebibliography}{99}
%\cite{Frixione:2006gn}
\bibitem{Frixione:2006gn}
  S.~Frixione and B.~R.~Webber,
  %``The MC@NLO 3.3 event generator,''
  hep-ph/0612272.
  %%CITATION = HEP-PH/0612272;%%

  %\cite{Frixione:2007vw}
  \bibitem{Frixione:2007vw}
  S.~Frixione, P.~Nason and C.~Oleari,
  %``Matching NLO QCD computations with Parton Shower simulations: the POWHEG
  %method,''
  JHEP {\bf 0711} (2007) 070
  [arXiv:0709.2092 [hep-ph]].
  %%CITATION = JHEPA,0711,070;%%
  %\cite{Alioli:2008gx}
\bibitem{Alioli:2008gx}
  S.~Alioli, P.~Nason, C.~Oleari and E.~Re,
%  %``NLO vector-boson production matched with shower in POWHEG,''
  JHEP {\bf 0807} (2008) 060
  [arXiv:0805.4802 [hep-ph]].
  %%CITATION = JHEPA,0807,060;%%
%\cite{Arnold:1990yk}
\bibitem{Arnold:1990yk}
  P.~B.~Arnold and R.~P.~Kauffman,
  %``W and Z production at next-to-leading order: From large q(t) to small,''
  Nucl.\ Phys.\  B {\bf 349} (1991) 381;
  %%CITATION = NUPHA,B349,381;%%
%
%\cite{Balazs:1995nz}
%\bibitem{Balazs:1995nz}
  C.~Balazs, J.~w.~Qiu and C.~P.~Yuan,
  %``Effects of QCD resummation on distributions of leptons from the decay of
  %electroweak vector bosons,''
  Phys.\ Lett.\  B {\bf 355} (1995) 548
  [hep-ph/9505203];
  %%CITATION = PHLTA,B355,548;%%
%
%%\cite{Balazs:1997xd}
%\bibitem{Balazs:1997xd}
  C.~Balazs and C.~P.~Yuan,
  %``Soft gluon effects on lepton pairs at hadron colliders,''
  Phys.\ Rev.\  D {\bf 56} (1997) 5558
  [hep-ph/9704258];
%
%\cite{Ellis:1997sc}
%\bibitem{Ellis:1997sc}
  R.~K.~Ellis, D.~A.~Ross and S.~Veseli,
  %``Vector boson production in hadronic collisions,''
  Nucl.\ Phys.\  B {\bf 503} (1997) 309
  [hep-ph/9704239];
  %%CITATION = NUPHA,B503,309;%%
%
%\cite{Ellis:1997ii}
%\bibitem{Ellis:1997ii}
  R.~K.~Ellis and S.~Veseli,
  %``W and Z transverse momentum distributions: Resummation in q(T)-space,''
  Nucl.\ Phys.\  B {\bf 511} (1998) 649
  [hep-ph/9706526];
  %%CITATION = NUPHA,B511,649;%%
%
%\cite{Qiu:2000ga}
%\bibitem{Qiu:2000ga}
  J.~w.~Qiu and X.~f.~Zhang,
  %``QCD prediction for heavy boson transverse momentum distributions,''
  Phys.\ Rev.\ Lett.\  {\bf 86} (2001) 2724
  [hep-ph/0012058] and
  %%CITATION = PRLTA,86,2724;%%
%
%\cite{Qiu:2000hf}
%\bibitem{Qiu:2000hf}
  J.~w.~Qiu and X.~f.~Zhang,
  %``Role of the nonperturbative input in QCD resummed Drell-Yan
  %$Q_T$-distributions,''
  Phys.\ Rev.\  D {\bf 63} (2001) 114011
  [hep-ph/0012348];
  %%CITATION = PHRVA,D63,114011;%%
%
%\cite{Kulesza:2001jc}
%\bibitem{Kulesza:2001jc}
  A.~Kulesza and W.~J.~Stirling,
  %``Soft gluon resummation in transverse momentum space for electroweak  boson
  %production at hadron colliders,''
  Eur.\ Phys.\ J.\  C {\bf 20} (2001) 349
  [hep-ph/0103089];
  %%CITATION = EPHJA,C20,349;%%
%
%\cite{Kulesza:2002rh}
%\bibitem{Kulesza:2002rh}
  A.~Kulesza, G.~Sterman and W.~Vogelsang,
  %``Joint resummation in electroweak boson production,''
  Phys.\ Rev.\  D {\bf 66} (2002) 014011
  [hep-ph/0202251];
  %%CITATION = PHRVA,D66,014011;%%
%
%\cite{Landry:2002ix}
%\bibitem{Landry:2002ix}
  F.~Landry, R.~Brock, P.~M.~Nadolsky and C.~P.~Yuan,
  %``Tevatron Run-1 Z boson data and Collins-Soper-Sterman resummation
  %formalism,''
  Phys.\ Rev.\  D {\bf 67} (2003) 073016
  [hep-ph/0212159];
  %%CITATION = PHRVA,D67,073016;%%
%
%\cite{Berge:2005rv}
%\bibitem{Berge:2005rv}
  S.~Berge, P.~M.~Nadolsky and F.~I.~Olness,
  %``Heavy-flavor effects in soft gluon resummation for electroweak boson
  %production at hadron colliders,''
  Phys.\ Rev.\  D {\bf 73} (2006) 013002
  [hep-ph/0509023];
  %%CITATION = PHRVA,D73,013002;%%
%
%\cite{Bozzi:2008bb}
%\bibitem{Bozzi:2008bb}
%  G.~Bozzi, S.~Catani, G.~Ferrera, D.~de Florian and M.~Grazzini,
  G.~Bozzi {\it et al.},
  %``Transverse-momentum resummation: a perturbative study of Z production at
  %the Tevatron,''
  Nucl.\ Phys.\  B {\bf 815} (2009) 174
  [arXiv:0812.2862 [hep-ph]].
  %%CITATION = NUPHA,B815,174;%%

%\cite{Bozzi:2010xn}
\bibitem{Bozzi:2010xn}
  G.~Bozzi, S.~Catani, G.~Ferrera, D.~de Florian, M.~Grazzini,
  %``Production of Drell-Yan lepton pairs in hadron collisions:
%Transverse-momentum resummation at next-to-next-to-leading logarithmic
%accuracy,''
  Phys.\ Lett.\  {\bf B696 } (2011)  207-213.
  [arXiv:1007.2351 [hep-ph]].


%\cite{Melnikov:2006di}
\bibitem{Melnikov:2006di}
  K.~Melnikov and F.~Petriello,
  %``The W boson production cross section at the LHC through O(alpha(s)**2),''
  Phys.\ Rev.\ Lett.\  {\bf 96} (2006) 231803
  [hep-ph/0603182] and
  %%CITATION = PRLTA,96,231803;%%
%\cite{Melnikov:2006kv}
%\bibitem{Melnikov:2006kv}
%  K.~Melnikov and F.~Petriello,
  %``Electroweak gauge boson production at hadron colliders through
  %O(alpha(s)**2),''
  Phys.\ Rev.\  D {\bf 74} (2006) 114017
  [hep-ph/0609070];  
K.~Melnikov and F.~Petriello,
  %``The W boson production cross section at the LHC through O(alpha(s)**2),''
  Phys.\ Rev.\ Lett.\  {\bf 96} (2006) 231803
  [hep-ph/0603182] and
  %%CITATION = PRLTA,96,231803;%%
%\cite{Melnikov:2006kv}
%\bibitem{Melnikov:2006kv}
%  K.~Melnikov and F.~Petriello,
  %``Electroweak gauge boson production at hadron colliders through
  %O(alpha(s)**2),''
  Phys.\ Rev.\  D {\bf 74} (2006) 114017
  [hep-ph/0609070].
  %%CITATION = PHRVA,D74,114017;%%
%\cite{Catani:2009sm}
\bibitem{Catani:2009sm}
  S.~Catani, L.~Cieri, G.~Ferrera, D.~de Florian and M.~Grazzini,
  %``Vector boson production at hadron colliders: a fully exclusive QCD
  %calculation at NNLO,''
  Phys.\ Rev.\ Lett.\  {\bf 103} (2009) 082001
  [arXiv:0903.2120 [hep-ph]].
%\cite{Gavin:2010az}
\bibitem{Gavin:2010az}
  R.~Gavin, Y.~Li, F.~Petriello, S.~Quackenbush,
  %``FEWZ 2.0: A code for hadronic Z production at next-to-next-to-leading
%order,''
  Comput.\ Phys.\ Commun.\  {\bf 182 } (2011)  2388-2403.
  [arXiv:1011.3540 [hep-ph]].
%\cite{Moch:2005ky}
\bibitem{Moch:2005ky}
  S.~Moch and A.~Vogt,
  %``Higher-order soft corrections to lepton pair and Higgs boson  production,''
  Phys.\ Lett.\  B {\bf 631} (2005) 48
  [hep-ph/0508265];
  %%CITATION = PHLTA,B631,48;%%
%\cite{Laenen:2005uz}
%\bibitem{Laenen:2005uz}
  E.~Laenen and L.~Magnea,
  %``Threshold resummation for electroweak annihilation from DIS data,''
  Phys.\ Lett.\  B {\bf 632} (2006) 270 
  [hep-ph/0508284];
  %%CITATION = PHLTA,B632,270;%%
%\cite{Idilbi:2005ni}
%\bibitem{Idilbi:2005ni}
  A.~Idilbi, X.~Ji, J.~P.~Ma and F.~Yuan,
  %``Threshold resummation for Higgs production in effective field theory,''
  Phys.\ Rev.\  D {\bf 73} (2006) 077501 
  [hep-ph/0509294];
  %%CITATION = PHRVA,D73,077501;%%
%\cite{Ravindran:2007sv}
%\bibitem{Ravindran:2007sv}
  V.~Ravindran and J.~Smith,
  %``Threshold corrections to rapidity distributions of Z and W^\pm bosons
  %beyond N^2 LO at hadron colliders,''
  Phys.\ Rev.\  D {\bf 76} (2007) 114004 
  [arXiv:0708.1689 [hep-ph]].
  %%CITATION = PHRVA,D76,114004;%%

\bibitem{Zykunov:2001mn}
  V.~A.~Zykunov,
  %``Electroweak corrections to the observables of W-boson production at
  %RHIC,''
  Eur.\ Phys.\ J.\ direct C {\bf 3} (2001) 9
  [hep-ph/0107059].
  %%CITATION = EPHJD,C3,9;%%

\bibitem{Dittmaier:2001ay}
  S.~Dittmaier and M.~Kr\"amer,
  % ``Electroweak radiative corrections to W-boson production at hadron
  %colliders,''
  Phys.\ Rev.\ D {\bf 65} (2002) 073007
  [hep-ph/0109062].
  %%CITATION = HEP-PH 0109062;%%

\bibitem{Baur:2004ig}
  U.~Baur and D.~Wackeroth,
  %``Electroweak radiative corrections to p (anti-)p --> W+- --> l+- nu  beyond
  %the pole approximation,''
  Phys.\ Rev.\  D {\bf 70} (2004) 073015
  [hep-ph/0405191];
  A.~Arbuzov, D.~Bardin, S.~Bondarenko, P.~Christova, L.~Kalinovskaya,
  G.~Nanava and R.~Sadykov,
  %``One-loop corrections to the Drell-Yan process in SANC. I: The charged
  %current case,''
  Eur.\ Phys.\ J.\  C {\bf 46} (2006) 407
  [Erratum-ibid.\  C {\bf 50} (2007) 505]
  [hep-ph/0506110].
  %%CITATION = EPHJA,C46,407;%%

\bibitem{CarloniCalame:2006zq}
  C.~M.~Carloni Calame, G.~Montagna, O.~Nicrosini and A.~Vicini,
  %``Precision electroweak calculation of the charged current Drell-Yan
  %process,''
  JHEP {\bf 0612} (2006) 016
  [hep-ph/0609170].
  %%CITATION = JHEPA,0612,016;%%

\bibitem{Brensing:2007qm}
  S.~Brensing, S.~Dittmaier, M.~Kr\"amer and A.~M\"uck,
  %``Radiative corrections to $W^-$ boson hadroproduction: Higher-order
  %electroweak and supersymmetric effects,''
  Phys.\ Rev.\  D {\bf 77} (2008) 073006
  [arXiv:0710.3309 [hep-ph]].
  %%CITATION = PHRVA,D77,073006;%%

%\cite{Baur:1997wa}
\bibitem{Baur:1997wa}
  U.~Baur, S.~Keller and W.~K.~Sakumoto,
  %``QED radiative corrections to Z boson production and the forward  backward
  %asymmetry at hadron colliders,''
  Phys.\ Rev.\  D {\bf 57} (1998) 199
  [hep-ph/9707301];
  %%CITATION = PHRVA,D57,199;%%
%
%\cite{Baur:2001ze}
%\bibitem{Baur:2001ze}
%  U.~Baur, O.~Brein, W.~Hollik, C.~Schappacher and D.~Wackeroth,
  U.~Baur {\it et al.},
  %``Electroweak radiative corrections to neutral-current Drell-Yan  processes
  %at hadron colliders,''
  Phys.\ Rev.\  D {\bf 65} (2002) 033007
  [hep-ph/0108274];
  %%CITATION = PHRVA,D65,033007;%%
%
%\cite{Zykunov:2005tc}
%\bibitem{Zykunov:2005tc}
  V.~A.~Zykunov,
  %``Weak radiative corrections to the Drell-Yan process for large invariant
  %mass of a dilepton pair,''
  Phys.\ Rev.\  D {\bf 75} (2007) 073019
  [hep-ph/0509315];
  %%CITATION = PHRVA,D75,073019;%%
%
%\cite{Arbuzov:2007db}
%\bibitem{Arbuzov:2007db}
%  A.~Arbuzov, D.~Bardin, S.~Bondarenko, P.~Christova, L.~Kalinovskaya,
G.~Nanava and R.~Sadykov,
  A.~Arbuzov {\it et al.},
  %``One-loop corrections to the Drell--Yan process in SANC (II). The neutral
  %current case,''
  Eur.\ Phys.\ J.\  C {\bf 54} (2008) 451
  [arXiv:0711.0625 [hep-ph]].
  %%CITATION = EPHJA,C54,451;%%

%\cite{CarloniCalame:2007cd}
\bibitem{CarloniCalame:2007cd}
  C.~M.~Carloni Calame, G.~Montagna, O.~Nicrosini and A.~Vicini,
  %``Precision electroweak calculation of the production of a high
  %transverse-momentum lepton pair at hadron colliders,''
  JHEP {\bf 0710} (2007) 109
  [arXiv:0710.1722 [hep-ph]].
  %%CITATION = JHEPA,0710,109;%%
%

%\cite{Dittmaier:2009cr}
\bibitem{Dittmaier:2009cr}
  S.~Dittmaier and M.~Huber,
  %``Radiative corrections to the neutral-current Drell-Yan process in the
%Standard Model and its minimal supersymmetric extension,''
  JHEP {\bf 1001} (2010)  060.
  [arXiv:0911.2329 [hep-ph]].
%\cite{CarloniCalame:2003ux}
\bibitem{CarloniCalame:2003ux}
  C.~M.~Carloni Calame, G.~Montagna, O.~Nicrosini and M.~Treccani,
  %``Higher-order QED corrections to W-boson mass determination at hadron
  %colliders,''
  Phys.\ Rev.\  D {\bf 69} (2004) 037301
  [hep-ph/0303102].
  %%CITATION = PHRVA,D69,037301;%%

%\cite{Placzek:2003zg}
\bibitem{Placzek:2003zg}
  W.~Placzek and S.~Jadach,
  %``Multiphoton radiation in leptonic W-boson decays,''
  Eur.\ Phys.\ J.\  C {\bf 29} (2003) 325
  [hep-ph/0302065].
  %%CITATION = EPHJA,C29,325;%%
%\cite{Carloni Calame:2004qw}
\bibitem{Carloni Calame:2004qw}
  C.~M.~Carloni Calame, S.~Jadach, G.~Montagna, O.~Nicrosini, W.~Placzek,
  %``Comparisons of the Monte Carlo programs HORACE and WINHAC for single W
% boson production at hadron colliders,''
  Acta Phys.\ Polon.\  {\bf B35 } (2004)  1643-1674.
  [hep-ph/0402235].

%\cite{Kilgore:2011pa}
\bibitem{Kilgore:2011pa}
  W.~B.~Kilgore, C.~Sturm,
  %``Two-Loop Virtual Corrections to Drell-Yan Production at order $\alpha_s
%\alpha^3$,''
   [arXiv:1107.4798 [hep-ph]];
%\cite{Kotikov:2007vr}
%\bibitem{Kotikov:2007vr}
  A.~Kotikov, J.~H.~K\"uhn, O.~Veretin,
  %``Two-Loop Formfactors in Theories with Mass Gap and Z-Boson Production,''
  Nucl.\ Phys.\  {\bf B788 } (2008)  47-62.
  [hep-ph/0703013 [HEP-PH]];
%\bibitem{Cao:2004yy}
  Q.~H.~Cao and C.~P.~Yuan,
  %``Combined effect of QCD resummation and QED radiative correction to W  boson
  %observables at the Tevatron,''
  Phys.\ Rev.\ Lett.\  {\bf 93} (2004) 042001
  [hep-ph/0401026];
  %%CITATION = PRLTA,93,042001;%%
  B.~F.~L.~Ward, C.~Glosser, S.~Jadach and S.~A.~Yost,
  %``Threshold corrections in precision LHC physics: QED x QCD,''
  Int.\ J.\ Mod.\ Phys.\  A {\bf 20} (2005) 3735
  [hep-ph/0411047];
  %%CITATION = IMPAE,A20,3735;%%
  B.~F.~L.~Ward and S.~A.~Yost,
  %``QED x QCD Resummation and Shower/ME Matching for LHC Physics,''
  Acta Phys.\ Polon.\  B {\bf 38} (2007) 2395
  [arXiv:0704.0294 [hep-ph]];
  %%CITATION = APPOA,B38,2395;%%
%  G.~Montagna, talk given at ``Loopfest VI: Radiative Corrections for the LHC
%and ILC'', Fermilab, 2007.
%\cite{Vicini:2007zz}
%\bibitem{Vicini:2007zz}
  A.~Vicini {\it et al.},
  %``Drell-Yan processes at the LHC,''
  PoS {\bf RADCOR2007} (2007) 013;
  %%CITATION = POSCI,RADCOR2007,013;%%
%\cite{Balossini:2008zz}
%\bibitem{Balossini:2008zz}
  G.~Balossini {\it et al.},
  %``Drell-Yan processes at hadron colliders,''
  Nuovo Cim.\  {\bf 123B} (2008) 741;
  %%CITATION = NUCIA,123B,741;%%
%\cite{Denner:2010vi}
  A.~Denner, S.~Dittmaier, T.~Kasprzik, and A.~M\"uck,
  %``Electroweak precision for W+jet production,''
  PoS {\bf RADCOR2009}, 058 (2010)
  [arXiv:1001.2468 [hep-ph]].
\bibitem{catani:ppgaga}
S.~Catani {\it et al.}, arXiv:1110.2375 [hep-ph].
%\cite{Nason:2006hfa}
\bibitem{Nason:2006hfa}
  P.~Nason, G.~Ridolfi,
  %``A Positive-weight next-to-leading-order Monte Carlo for Z pair
%hadroproduction,''
  JHEP {\bf 0608 } (2006)  077.
  [hep-ph/0606275].
%\cite{Binoth:2006mf}
\bibitem{Binoth:2006mf}
  T.~Binoth, M.~Ciccolini, N.~Kauer, M.~Kramer,
  %``Gluon-induced W-boson pair production at the LHC,''
  JHEP {\bf 0612 } (2006)  046.
  [hep-ph/0611170].

%\cite{Accomando:2001fn}
\bibitem{Accomando:2001fn}
  E.~Accomando, A.~Denner, S.~Pozzorini,
  %``Electroweak correction effects in gauge boson pair production at the CERN
%LHC,''
  Phys.\ Rev.\  {\bf D65 } (2002)  073003.
  [hep-ph/0110114];
%\cite{Accomando:2004de}
%\bibitem{Accomando:2004de}
  E.~Accomando, A.~Denner, A.~Kaiser,
  %``Logarithmic electroweak corrections to gauge-boson pair production at the
%LHC,''
  Nucl.\ Phys.\  {\bf B706 } (2005)  325-371.
  [hep-ph/0409247];
%\cite{Accomando:2005ra}
%\bibitem{Accomando:2005ra}
  E.~Accomando, A.~Denner, C.~Meier,
  %``Electroweak corrections to $W \gamma$ and $Z \gamma$ production at the
%LHC,''
  Eur.\ Phys.\ J.\  {\bf C47 } (2006)  125-146.
  [hep-ph/0509234].
\bibitem{Bierweiler}
A.~Bierweiler, T.~Kasprzik, J.~H.~K\"uhn, S.~Uccirati, in preparation.
%\cite{Kuhn:2011mh}
\bibitem{Kuhn:2011mh}
  J.~H.~K\"uhn, F.~Metzler, A.~A.~Penin, S.~Uccirati,
  %``Next-to-Next-to-Leading Electroweak Logarithms for W-Pair Production at
%LHC,''
  JHEP {\bf 1106 } (2011)  143.
  [arXiv:1101.2563 [hep-ph]].

%%\cite{Giele:1993dj}
%\bibitem{Giele:1993dj}
%  W.~T.~Giele, E.~W.~N.~Glover and D.~A.~Kosower,
%  %``Higher order corrections to jet cross-sections in hadron colliders,''
%  Nucl.\ Phys.\  B {\bf 403} (1993) 633
%  [hep-ph/9302225].
%  %%CITATION = NUPHA,B403,633;%%

\bibitem{Alioli:2010qp}
  S.~Alioli, P.~Nason, C.~Oleari, E.~Re,
  %``Vector boson plus one jet production in POWHEG,''
  JHEP {\bf 1101 } (2011)  095.
  [arXiv:1009.5594 [hep-ph]].   
 %\cite{Campbell:2002tg}

\bibitem{Campbell:2002tg}
  J.~M.~Campbell and R.~K.~Ellis,
  %``Next-to-leading order corrections to $W^+$ 2 jet and $Z^+$ 2 jet production
  %at hadron colliders,''
  Phys.\ Rev.\  D {\bf 65} (2002) 113007
  [hep-ph/0202176].
  %%CITATION = PHRVA,D65,113007;%% 

%%\cite{vanderBij:1988ac}
%\bibitem{vanderBij:1988ac}
% J.~J.~van der Bij, E.~W.~N.~Glover,
% %``Z Boson Production And Decay Via Gluons,''
% Nucl.\ Phys.\  {\bf B313 } (1989)  237. 



  %%CITATION = ARXIV:1009.2338;%%
%\cite{Berger:2010vm}
\bibitem{Berger:2010vm}
  C.~F.~Berger {\it et al.},
  %``Next-to-Leading Order QCD Predictions for Z,gamma^*+3-Jet Distributions
  %at the Tevatron,''
  Phys.\ Rev.\  {\bf D82 } (2010)  074002.
  [arXiv:1004.1659 [hep-ph]];
%\cite{Berger:2009ep}
%\bibitem{Berger:2009ep}
  C.~F.~Berger {\it et al.},
  %``Next-to-Leading Order QCD Predictions for W+3-Jet Distributions at Hadron
  %Colliders,''
  Phys.\ Rev.\  D {\bf 80} (2009) 074036
  [arXiv:0907.1984 [hep-ph]];
  %%CITATION = PHRVA,D80,074036;%%
%\cite{Ellis:2008qc}
%\bibitem{Ellis:2008qc}
  R.~K.~Ellis, W.~T.~Giele, Z.~Kunszt, K.~Melnikov and G.~Zanderighi,
  %``One-loop amplitudes for $W^+$ 3 jet production in hadron collisions,''
  JHEP {\bf 0901}, 012 (2009)
  [arXiv:0810.2762 [hep-ph]].
  %%CITATION = JHEPA,0901,012;%%
%\cite{Alioli:2010qp}

%\cite{Berger:2010zx}
\bibitem{Berger:2010zx}
  C.~F.~Berger {\it et al.},
  %``Precise Predictions for W + 4 Jet Production at the Large Hadron
  %Collider,''
  arXiv:1009.2338 [hep-ph].

%\cite{Kuhn:2004em}
\bibitem{Kuhn:2004em}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``Logarithmic electroweak corrections to hadronic Z+1 jet production at large
%transverse momentum,''
  Phys.\ Lett.\  {\bf B609 } (2005)  277-285.
  [hep-ph/0408308];
%\cite{Kuhn:2005az}
%\bibitem{Kuhn:2005az}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``One-loop weak corrections to hadronic production of Z bosons at large
transverse momenta,''
  Nucl.\ Phys.\  {\bf B727 } (2005)  368-394.
  [hep-ph/0507178];
%\bibitem{Kuhn:2007qc}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``Electroweak corrections to large transverse momentum production of W bosons
  %at the LHC,''
  Phys.\ Lett.\  B {\bf 651} (2007) 160
  [hep-ph/0703283];
  %%CITATION = PHLTA,B651,160;%%
%
%\bibitem{Hollik:2007sq}
  W.~Hollik, T.~Kasprzik and B.~A.~Kniehl,
  %``Electroweak corrections to W-boson hadroproduction at finite transverse
  %momentum,''
  Nucl.\ Phys.\  B {\bf 790} (2008) 138
  [arXiv:0707.2553 [hep-ph]].
  %%CITATION = ARXIV:0707.2553;%%
%\cite{Kuhn:2007cv}
\bibitem{Kuhn:2007cv}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``Electroweak corrections to hadronic production of W bosons at large
  %transverse momenta,''
  Nucl.\ Phys.\  B {\bf 797} (2008) 27
  [arXiv:0708.0476 [hep-ph]].
  %%CITATION = ARXIV:0708.0476;%%
%\cite{Denner:2009gj}
\bibitem{Denner:2009gj}
  A.~Denner, S.~Dittmaier, T.~Kasprzik, and A.~M\"uck,
  %``Electroweak corrections to W + jet hadroproduction including leptonic
%W-boson decays,''
  JHEP {\bf 0908 } (2009)  075.
  [arXiv:0906.1656 [hep-ph]];
%\cite{Denner:2011vu}
%\bibitem{Denner:2011vu}
  A.~Denner, S.~Dittmaier, T.~Kasprzik, A.~M\"uck,
  %``Electroweak corrections to dilepton + jet production at hadron colliders,''
  JHEP {\bf 1106 } (2011)  069.
  [arXiv:1103.0914 [hep-ph]].
\samepage

\end{thebibliography}


%\section{Introduction}
\section{The Drell--Yan process}\noindent
A proper understanding of the charged-current (CC) and neutral-current
(NC) single vector-boson production process $\mathrm{pp} \to
\PW^{\pm}/\PZ,\gamma^* \to l\nu_l / l\bar{l}$ is an important task at
the LHC. Such processes have a clean leptonic signature and are
therefore well suited for luminosity monitoring and
detector calibration purposes. Moreover, Z-boson production will allow
for a precise determination of the effective weak mixing angle at the
LHC by measuring appropriate forward-backward asymmetries, while W-boson
production is primarily used to precisely determine the W-boson mass
$M_{\PW}$ and width $\Gamma_{\PW}$, and to constrain the parton
distribution functions (PDFs) by analyzing the $\PW^-/\PW^+$-ratio or
the W-boson charge asymmetry $A_{\PW}^l \equiv (\rd \sigma_{l^+}/\rd
\eta_l -\rd \sigma_{l^-}/\rd \eta_l)/(\rd \sigma_{l^+}/\rd \eta_l +\rd
\sigma_{l^-}/\rd \eta_l)$, respectively.
\begin{figure}[]
\begin{center}
%\includegraphics[width = 0.5\textwidth]{cmp1-z-lhc-MCatNLO-POWHEG}
%\put(-90,-5){\tiny \textcolor{cite}{[Alioli et al.: arXiv:0805.4802 [hep-ph]]}}
\includegraphics[width = 0.45\textwidth]{nnllnlo.eps}
%\put(-90,-5){\tiny \textcolor{cite}{[Bozzi et al.: arXiv:1007.2351v2
%[hep-ph]]}}
\includegraphics[width = 0.5\textwidth]{HPDFcomp_ZpT.eps}
%\put(-90,-5){\cita{Gavin et al.: arXiv:1011.3540}}
%\caption{Left: Different parton-shower implementations of single-Z
%  production at the LHC~\cite{Alioli:2008gx}; Right: Comparison of the
%  resummed result NNLL accuracy to the
%  fixed-order computation at the Tevatron~\cite{Bozzi:2010xn}. The
%  fixed-order prediction fails for $q_{\rT} < 10\,
%  \mathrm{GeV}$. ($\mathrm{(fin.)_{NLO}}$: Non-logarithmically-enhanced
%  contributions)\label{fi:DYRes}}
\caption{Left: Comparison of the resummed result at NNLL accuracy to the
  fixed-order computation at the Tevatron~\cite{Bozzi:2010xn} for the
  Drell--Yan process. The fixed-order prediction fails for $q_{\rT} <
  10\, \mathrm{GeV}$. ($\mathrm{(fin.)_{NLO}}$:
  Non-logarithmically-enhanced contributions)\label{fi:DYRes}. Right:
  Distribution of the Z-boson transverse momentum at NNLO accuracy for
  different PDF sets at the LHC ({\tt FEWZ}~\cite{Gavin:2010az}).}
\end{center}
\end{figure}

At hadron colliders, at least the knowledge of the next-to-leading-order (NLO)
QCD corrections is crucial for the normalization of the total cross section,
the reduction of the scale dependence, and to obtain realistic predictions for
the shapes of differential cross sections. The QCD corrections to the
Drell--Yan (DY) process have been studied extensively. The NLO contributions
have been matched with parton
showers~\cite{Frixione:2006gn,Frixione:2007vw,Alioli:2008gx} and combined with
soft-gluon resummation~\cite{Arnold:1990yk, Bozzi:2010xn} to efficiently
account for dominating contributions $\propto \alps^n\ln^m(M_V^2/q_{\rT}^2)$
at small transverse momenta $q_{\rT}$ of the vector bosons (see
Fig.~\ref{fi:DYRes}, left). Moreover, the corresponding
next-to-next-to-leading-order (NNLO) two-loop corrections are known fully
differentially (see Fig.~\ref{fi:DYRes}, right) and have been implemented in
Monte Carlo programs~\cite{Melnikov:2006di, Catani:2009sm, Gavin:2010az}. Even
the $\mathrm{N^3LO}$ corrections are known in the soft-plus-virtual
approximation~\cite{Moch:2005ky}, pushing the theoretical uncertainties due to
perturbative QCD to the level of below 2\% for this specific process class.
%\begin{figure}[]
%\begin{center}
%\includegraphics[width = 0.5\textwidth]{HPDFcomp_ZpT.ps}
%%\put(-90,-5){\cita{Gavin et al.: arXiv:1011.3540}}
%\includegraphics[width = 0.45\textwidth]{y34lhc}
%%\put(-90,-5){\tiny \textcolor{cite}{[Catani et al: arXiv:0903.2120
%%[hep-ph]]}}
%\caption{Left: Distribution of the Z-boson transverse momentum at NNLO
%  accuracy for different PDF sets ({\tt FEWZ}~\cite{Gavin:2010az}); Right: NNLO
%  prediction for the Z-boson rapidity
%  distribution ({\tt DYNNLO}~\cite{Catani:2009sm}).\label{fi:DYNNLO}}
%\end{center}
%\end{figure}

The corresponding NLO electroweak (EW) corrections have been
investigated by many groups for the CC~\cite{Zykunov:2001mn,
  Dittmaier:2001ay, Baur:2004ig, CarloniCalame:2006zq, Brensing:2007qm}
and the NC~\cite{Baur:1997wa,CarloniCalame:2007cd,Dittmaier:2009cr}
process, and tuned comparisons of different implementations have been
performed.  Universal higher-order
corrections were included~\cite{CarloniCalame:2003ux}, and the
predictions were studied in different input schemes, where corrections 
due to $\Delta \alpha$ and $\Delta \rho$ are absorbed in effective
leading-order (LO) couplings. It was found that the relative EW
corrections are nearly insensitive to the specific theoretical treatment
of the vector-boson resonance and to the inclusion of virtual corrections
within the MSSM, respectively~\cite{Brensing:2007qm, Dittmaier:2009cr}.
%\begin{figure}
%  \centering
%  \includegraphics[width = 0.9\textwidth]{cmpinputschemes.epsi}
%\caption{dhfsjkhf fd sfsd fdsf sdf sd f s d f sd fsdfsdf s d fsdf s dfs
%  df sdf sd fsd f sd fsd fsdf sdf  sdfsdf sdfsdf s d}
%%\put(-100,5){\tiny \textcolor{cite}{[Dittmaier, Huber: arXiv:0911.2329v2
%[hep-ph]]}}
%\end{figure}

In general, the EW corrections at moderate energies are dominated by
final-state photon radiation off leptons leading to a significant
distortion of the line-shapes of the leptonic invariant-mass and
transverse-momentum distributions (see Fig.~\ref{fi:EWCC}), which
strongly influences the precise determination of
$M_{\PW}$. Consequently, also the effect of multi-photon radiation has
been investigated in a structure-function
approach~\cite{Placzek:2003zg}, and the corresponding contributions have
been matched to the fixed-order $\mathcal{O}(\alpha)$ corrections within
{\tt HORACE}~\cite{Carloni Calame:2004qw}.  Moreover, first steps have
been taken towards a combined QCD $\otimes$ EW (two-loop)
analysis~\cite{Kilgore:2011pa} to improve the EW accuracy to the level
of better than 1\%.
\begin{figure}[]
\begin{center}
\includegraphics[width = .95\textwidth]{pt_plot_new_cuts_EW}
%\put(-90,-10){\tiny \textcolor{cite}{[Brensing et al.:
%arXiv:0710.3309v2 [hep-ph]]}}
\caption{EW corrections to the lepton transverse momentum for the CC
  Drell--Yan process at the LHC~\cite{Brensing:2007qm}. Results are
  presented for bare muons ($\delta_{q\bar{q}}^{\mu^+\nu_{\mu}}$) and
  for electrons employing electron-photon
  recombination ($\delta^{\mathrm{rec}}_{q\bar{q}}$), respectively; the
  effects due to multi-photon radiation
  ($\delta_{\mathrm{multi}-\gamma}$) as well as photon-induced processes
  ($\delta_{q\gamma}$) are small.\label{fi:EWCC}}
\end{center}
\end{figure}
\section{Vector-boson pair production}\noindent
\begin{figure}[]
\begin{center}
\includegraphics[width = 0.95\textwidth]{plot_ww_vergl}
\caption{W-boson pair production at the LHC at 14 TeV; Left: Comparison 
  of the full LO cross section to the high-energy approximation for
  different values of $p_{\rT,\PW}^{\mathrm{cut}}$. Right: Same for the
  relative corrections (Approximate NLO corrections at NNLL accuracy). For
  $p_{\rT,\PW}^{\mathrm{cut}} < 1\,\mathrm{TeV}$, the approximate results
  become crude~\cite{Bierweiler}.\label{fi:Wpair}}
\end{center}
\end{figure}
Vector-boson pair production processes $\mathrm{pp} \to V_1V_2 \to 4 l$
($V_i = \PW^{\pm},\PZ,\gamma$) play a key role in the understanding of
irreducible backgrounds to Standard-Model (SM) Higgs production in the
intermediate mass range. Moreover, they will allow us to probe the
non-abelian structure of the SM at high scales, and may give hints to
the existence of anomalous trilinear and quartic couplings, which are
predicted to have a sizable effect at high energies accessible at the
LHC. In addition, such processes constitute backgrounds to various
new-physics signatures with leptons and missing transverse energy.

The NLO QCD corrections to vector-boson pair production are known for a
long time, and a fully-exclusive computation of the two-loop corrections
to $\mathrm{pp} \to \gamma\gamma$ has been completed
recently~\cite{catani:ppgaga}. The NLO corrections have been matched
with parton showers and combined with soft-gluon
resummation~\cite{Frixione:2006gn,Nason:2006hfa}, and the leptonic
decays are accounted for in the narrow-width approximation, retaining
all spin information. Although the corresponding cross sections are
dominated by the $q\bar{q}$ channels, there are significant
contributions from the loop-induced channels $\Pg\Pg \to V_1V_2$ at the
LHC, especially if selection cuts for Higgs searches are
applied, of up to 30\%~\cite{Binoth:2006mf}.

In the high-energy limit the corrections at $\Or(\alpha)$ are known for
all combinations of external vector bosons in the pole
approximation~\cite{Accomando:2001fn} to facilitate a realistic
phenomenological description of the final-state leptons. Recently, we
have computed the full EW corrections to on-shell W-pair production at
the LHC~\cite{Bierweiler} and find sensible agreement with former
approximate results~\cite{Kuhn:2011mh} for center-off-mass energies
beyond 2~TeV (see Fig.~\ref{fi:Wpair}). For lower energies, however, the
approximation becomes crude even at leading order.  To this very
process, also two-loop pieces are available in NNLL
accuracy~\cite{Kuhn:2011mh}, leading to sizable positive corrections of
about 10\%.

\section{Vector-boson production with associated jets}\noindent
\begin{figure}
\begin{tabular}{cc}
%\begin{center}
\includegraphics[width=.45\textwidth]{Wm4jLHC7HTp_anti-kt-R5-Pt25_Allp_HT}
%\put(-90,-10){\tiny \textcolor{cite}{[Berger et al.: arXiv:1009.2338
%[hep-ph]]}} 
&
%\begin{center}
%\vspace{-1cm}
\includegraphics[width = .45\textwidth]{V+jet_os}
%\put(-70,-5){\tiny \textcolor{cite}{[K\"uhn et al.: arXiv:0708.0476
%    [hep-ph]]}}
%\end{center}
\end{tabular}
\caption{\label{fi:Wjets} Left: NLO QCD corrections to $\PW^{-}$ + 4
  jets production at the LHC~\cite{Berger:2010zx}; top: Differential
  distribution of the transverse energy $H_{\rT} \equiv \sum_j
  E_{\rT,j}^{\mathrm{jet}} + E_{\rT}^e + E_{\rT}^\nu$; bottom:
  $K$-factor and scale dependence; note the marked reduction of scale
  uncertainties at NLO. Right: Relative EW corrections to the
  $p_{\rT}$ distribution for on-shell $\PW^{\pm}$ + jet production
  at the LHC, and corresponding relative
  corrections~\cite{Kuhn:2007cv}. (NNLO = NLO + 2-loop NLL; NLL and NNLL
  considered at one loop)}
\end{figure}
\noindent
At the LHC, vector bosons are almost always produced together with one
or more hard jets. Since the corresponding cross sections are still
sizable, and due to the distinct leptonic final states, such processes
may help to attain a profound understanding of SM physics in a
hadron-collider environment.  Moreover, V + jet(s) production also
provides a good possibility to study QCD jet dynamics in great detail,
and the corresponding signatures with jets, charged leptons and missing
transverse energy have to be understood properly to discriminate
hypothetical new-physics signals from the SM background.

The NLO QCD corrections for W/Z + 1 jet have been matched with parton
showers~\cite{Alioli:2010qp}, and the NLO corrections for W/Z + 2 jets
are e.g.\ included in {\tt MCFM}~\cite{Campbell:2002tg}. NLO results for
\mbox{W/Z + 3 jets} have been presented by different
collaborations~\cite{Berger:2010vm}, and even the NLO results for
$\PW+4\;\mathrm{jets}$ could be worked out~\cite{Berger:2010zx}, using
unitarity-based techniques for the automatized calculation of virtual
amplitudes (see Fig.~\ref{fi:Wjets}).

Various EW effects to W/Z + 1 jet production have been investigated,
both in the on-shell~\cite{Kuhn:2004em,Kuhn:2007cv} and
off-shell~\cite{Denner:2009gj} scenario, where the leptonic decays of
the vector bosons as well as finite-width effects are fully taken into
account in the latter. As in the DY case, at high energies the
corrections are dominated by universal large logarithms $\propto
\alpha^L\ln^{2L-n}(M_V/\sqrt{s})$ ($\equiv$ N$^n$LL accuracy at $L$
loops) which are known up to two loops in the NLL approximation (see
Fig.~\ref{fi:Wjets}), while at low and medium energies the corrections
are nearly fully dominated by final-state photon radiation.

\section{Conclusions}\noindent
Precise theoretical predictions for vector-boson production at the LHC
have been established during the recent years, where both fixed-order
and resummed QCD and EW effects were considered. The accurate knowledge
of Drell--Yan physics will help to further constrain the PDFs and
probably enable the most accurate determination of $M_{\PW}$ with an
error of only 15 MeV at the LHC. Moreover, it will also open the door to a
fruitful analysis of EW precision observables at hadron
colliders. Apart from that, a profound theoretical knowledge of
vector-boson pair production and V + jet(s) is mandatory to properly
assess the SM backgrounds to Higgs-boson or $\mathrm{t\bar{t}}$
production, as well as to various beyond-SM physics scenarios.





\end{document}


